Author’s response to reviews

Title: Barriers and facilitators to implementing Dementia Care Mapping in Care Homes: results from the DCM™ EPIC Trial process evaluation

Authors:

Alys Griffiths (alys.griffiths@leedsbeckett.ac.uk)
Rachael Kelley (r.kelley@leedsbeckett.ac.uk)
Lucy Garrod (lucy.garrod@oxfordhealth.nhs.uk)
Devon Perfect (devon.perfect@oxfordhealth.nhs.uk)
Olivia Robinson (o.robinson@leedsbeckett.ac.uk)
Emily Shoesmith (e.shoesmith@leedsbeckett.ac.uk)
Joanne McDermid (joanne.mcdermid@leedsbeckett.ac.uk)
Natasha Burnley (n.burnley@leedsbeckett.ac.uk)
Claire Surr (c.a.surr@leedsbeckett.ac.uk)

Version: 3 Date: 18 Jan 2019

Author’s response to reviews:

1. We noticed that author Lucy Garrod is not listed on the title page of your manuscript, but is listed as an author in the submission system. Please be sure that the list of authors matches in both the submission system and the manuscript. Please note that if the omission was done in error, an authorship change form would not be required. However, as Lucy Garrod was listed as an author in the original submission and the revision, an authorship change form would need to be completed for removing them as an author.

Thank you for noticing this - I have re-added Lucy back into the authorship list, as she was removed in error.

2. Thank you for clarifying that the informed consent obtained was written. Please also state this in the Ethics section of your Declarations.

This has been added.
3. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

This has been added.

4. Please upload your figures as separate figure files and remove them from the main manuscript file. When uploading the figure files, please make sure they contain the image graphic only and not the titles or captions. Please place the titles and captions in the main manuscript after the References, under a “Figure Legends” heading.

These have now been uploaded as separate files.

5. Please add a “Background” heading at the beginning of the main text of your manuscript.

This has been added.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

This has been uploaded as a single clean version. Thank you.